

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Yamani, Amal; Hussainey, Khaled; Albitar, Khaldoon

Article Does governance affect compliance with IFRS 7?

Journal of Risk and Financial Management

Provided in Cooperation with: MDPI – Multidisciplinary Digital Publishing Institute, Basel

Suggested Citation: Yamani, Amal; Hussainey, Khaled; Albitar, Khaldoon (2021) : Does governance affect compliance with IFRS 7?, Journal of Risk and Financial Management, ISSN 1911-8074, MDPI, Basel, Vol. 14, Iss. 6, pp. 1-23, https://doi.org/10.3390/jrfm14060239

This Version is available at: https://hdl.handle.net/10419/239655

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Article **Does Governance Affect Compliance with IFRS 7?**

Amal Yamani¹, Khaled Hussainey^{2,*} and Khaldoon Albitar²

- ¹ Department of Accounting, Faculty of Economics and Administration, King Abdulaziz University, Jeddah 21589, Saudi Arabia; amyamani@kau.edu.sa
- ² Accounting and Financial Management Group, Faculty of Business and Law, University of Portsmouth, Portsmouth PO1 3DE, UK; Khaldoon.Albitar@port.ac.uk
- * Correspondence: Khaled.Hussainey@port.ac.uk; Tel.: +44-023-9284-4715

Abstract: Although there has been considerable research on the impact of corporate governance on corporate voluntary disclosure, empirical evidence on how governance affects compliance with mandatory disclosure requirements is limited. We contribute to governance and disclosure literature by examining the impact of corporate governance on compliance with IFRS 7 for the banking sector in Gulf Cooperation Council (GCC). We use a self-constructed disclosure index to measure compliance with IFRS 7. We use regression analyses to examine the impact of board characteristics, audit committee characteristics and ownership structure on compliance with IFRS 7. Using a sample of 335 bank-year observations for GCC listed banks over the period 2011–2017, we report evidence that corporate governance variables affect compliance with IFRS 7. However, the significance of these variables depends on the type of the regression model used. Our findings suggest that governance matters for mandatory disclosure requirements. So to improve the level of compliance, regulators, official authorities, and policymakers should intensify their efforts toward improving corporate governance codes, following up their implementation and enhancing the enforcement mechanisms.

Keywords: corporate governance; IFRS 7 compliance; GCC

1. Introduction

Prior research attributes the low level of compliance with International Financial Reporting Standards (IFRS) to the absence of adequate enforcement regulations (Alfaraih 2009; Al Mutawaa and Hewaidy 2010; Al-Shammari 2011; Alanezi and Albuloushi 2011; Tsalavoutas 2011; Al-Jabri and Hussain 2012; Bova and Pereira 2012; Santos et al. 2013). The literature also shows that corporate governance (CG) is a significant enforcement mechanism in IFRS implementation (Al-Akra et al. 2010; Abdul Rahman and Hamdan 2017; Al-Sartawi et al. 2016; Ebrahim and Fattah 2015; Juhmani 2017). In this paper, we are interested in IFRS 7 Financial Instruments: Disclosures. This standards requires banks to make several disclosures about the significance of financial instruments for the financial position and the financial performance of the entity concerned, and the nature and extend of risks to which the entity is exposed in relation to financial instruments, both in quantitative and qualitative terms (Melville 2017). The goal of the paper is to examine the impact of corporate governance mechanisms on the compliance with IFRS 7 standard. Our research question is: "Does corporate governance mechanisms affect the compliance levels of IFRS 7 standard by banks in Gulf Cooperation Council (GCC)"? Despite growing awareness of the significant role of governance in GCC, especially in improving disclosure and compliance with regulations, it is still in its early stages and faces challenges with enforcement mechanisms (Swedan and Ahmed 2019). Thus, our paper offers a novel contribution to governance and IFRS literature by providing the first empirical evidence on the impact of corporate governance mechanisms on the compliance with IFRS 7 standard.

We are motivated to focus on IFRS 7 in the GCC banking sector for several reasons. First, the six GCC countries are economically strong (oil-exporting countries) and have



Citation: Yamani, Amal, Khaled Hussainey, and Khaldoon Albitar. 2021. Does Governance Affect Compliance with IFRS 7? *Journal of Risk and Financial Management* 14: 239. https://doi.org/10.3390/ irfm14060239

Academic Editor: Ştefan Cristian Gherghina

Received: 28 April 2021 Accepted: 26 May 2021 Published: 28 May 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). cultural and political situations distinct from other countries. Second, the banking sector plays an important role in supporting the growth of diversified activities in the GCC region and it represents the largest sector relative to other sectors (Hamdan). Third, as in many other countries, the banking sector in GCC is more regulated than non-financial sectors and therefore is more likely to comply with disclosure requirements (Al Lawati and Hussainey 2021). Fourth, we focus on IFRS 7 as financial instruments are pervasive more in the banking sector. Prior research shows that financial instruments are one of the most important tools that the banking sector is dealing with (Allini et al. 2019).

The paper is structured as follows: Section 2 reviews the literature and develops the research hypotheses. Section 3 discusses the research design. Section 4 reports our empirical findings. Section 5 discusses the findings. Section 6 concludes.

2. Literature Review and Hypotheses Development

Corporate governance is defined as "the system by which companies are directed or controlled" (Huse 2007, p. 15). Corporate governance mechanisms have received considerable attention in accounting research and have been considered to be key drivers for IFRS adoption. A considerable number of papers focused on the impact of corporate governance on the level of compliance with IFRS in general (Al-Akra et al. 2010; Alanezi and Albuloushi 2011; Alfraih 2016; Almaqtari 2019; Bagudo et al. 2018; Juhmani 2017; Mbir et al. 2020) and not a single standard per se, and in particular IFRS 7. A few studies explored the drivers of financial instruments compliance. For example, Adznan and Nelson (2014) found that external and internal audit committee independence and audit fees have a positive impact on the level of MFRS 7compliance, while board attributes (board independence and board expertise) have no impact. Lopes and Rodrigues (2007) did not find an impact of board independence on the level of compliance with IAS 32 and IAS 39 in Portugal. In contrast, Tauringana and Chithambo (2016) found a positive effect of board independence on compliance with IFRS 7 in Malawi. Similar results are confirmed by Agyei-Mensah (2017) in that the level of compliance with IFRS 7 of 30 Ghanaian firms were affected by board independence and institutional ownership. Tahat et al. (2017) found that board independence and the existence of an audit committee have a positive impact on IFRS 7 compliance. Finally, Nahar et al. (2020) found that board size, independence and the existence of risk committees increase levels of mandatory risk disclosure. The current study complements this literature by examining the impact of corporate governance on IFRS 7 compliance in GCC.

One of the gaps of prior research on determinants of IFRS compliance is that most of these studies (Abdul Rahman and Hamdan 2017; Al-Akra et al. 2010; Alanezi and Albuloushi 2011; Alfraih 2016; Al-Sartawi et al. 2016; Bagudo et al. 2018; Bova and Pereira 2012; Ebrahim and Fattah 2015; Istiningrum 2020; Juhmani 2017; Mbir et al. 2020; Sanni et al. 2020; Sarea and Nesuf 2013; Uyar et al. 2016) test a single country to reach their results, which makes it difficult to generalize the results for other countries. Another gap is that limited corporate governance mechanisms have been used in the literature. For example, studies have concentrated on the board and ownership characteristics (Abdul Rahman and Hamdan 2017; Al-Akra et al. 2010; Alanezi and Albuloushi 2011; Alfraih and Almutawa 2017; Ebrahim and Fattah 2015), audit committee and ownership characteristics (Alanezi et al. 2012), board and audit committee characteristics (Adznan and Nelson 2014; Al-Sartawi et al. 2016), board characteristics (Alfraih 2016), and board, audit and ownership characteristics (Agyei-Mensah 2017; Juhmani 2017; Tahat et al. 2017). The current study fills these gaps.

This study examines the impact of several corporate governance mechanisms, including: board attributes (board size, board independence, board meeting frequency), the audit committee (AC) attributes (AC size, AC independence, AC meeting frequency), and ownership attributes (blockholder, governmental, institutional) and their effects on the degree of compliance.

• Board size

One of the fundamentals of corporate governance determinants is the size of the board of directors. Along with the many tasks of the board, it is an important link between management and the interests of the owners. In fact, an optimal number of directors has not yet been set. Literature has addressed the size of the board in two respects. The first trend is in favor of increasing the number of members on the board of directors. It assumes that a larger board leads to more diversity of knowledge and gives a greater chance of exchanging different experiences and information. It also helps to increase the effectiveness of the board's work and facilitates monitoring of performance related to compliance with regulations and standards. Accordingly, this interpretation is based on resource dependency theory (Al-Janadi et al. 2013; Almaqtari 2019). On the other hand, and from the perspective of agency theory, opponents argue that a larger board of directors would weaken the communication between members and may also slow the making of urgent decisions, which may weaken the effectiveness of the board's work and increase the costs of monitoring (Al-Akra et al. 2010; Bae et al. 2018).

The results of studies related to this factor are varied—some of which show a positive impact of board size on the degree of compliance with IFRS (Al-Akra et al. 2010; Alfraih 2016; Bagudo et al. 2018; Mnif and Znazen 2020), a negative impact (Alfraih and Almutawa 2017), while others show no significant impact on the degree of compliance with IFRS (Agyei-Mensah 2017; Almaqtari 2019; Ebrahim and Fattah 2015; Juhmani 2017; Zango et al. 2015). Based on the above discussion, we hypothesize that:

Hypothesis 1 (H1). *There is an association between board size and the extent to which banks comply with IFRS 7 mandatory disclosure.*

Board independence

Good corporate governance may include the proper composition of the board of directors. That composition includes executives (no separation between owners and management) or non-executives/independent (separation between owners and management) (Abdul Rahman and Hamdan 2017; Ebrahim and Fattah 2015). The independence of the board of directors has been debated in terms of what independence concept is and how it affects the effectiveness and performance of management. Under the umbrella of agency theory, it is expected that the existence of independent members (outsiders) will improve the level of supervision over the management and help to make appropriate decisions and increase the interdependence between internal and external parties, thus reducing agency problems (Bae et al. 2018; Bagudo et al. 2018; Ebrahim and Fattah 2015; Juhmani 2017). In addition, the independence of directors can enhance, to some extent, the shareholders' confidence by increasing the transparency in corporate outputs (annual reports) and integrity in the preparation of such reports. This, in turn, enhances the quality of disclosure and the level of compliance with mandatory applicable standards requirements (Ebrahim and Fattah 2015; Mnif and Znazen 2020; Tahat et al. 2017; Tauringana and Chithambo 2016). Several studies provided a significant positive correlation between both the degree of compliance with IFRS and the independence of the board (Agyei-Mensah 2017; Almagtari 2019; Bagudo et al. 2018; Juhmani 2017; Mnif and Znazen 2020; Tahat et al. 2017; Tauringana and Chithambo 2016).

However, the other side of this factor is based on stewardship theory, which enhances the role of 'centralized' inside-directors to improve work effectiveness and performance. Since those directors are sufficiently aware of the company information and tasks and the performance of its employees, this consequently helps to reinforce their commitment toward regulations and to follow up on compliance with laws and standards. Moreover, such independence in the board may not be achieved fully 'in reality', and therefore may not lead to the achievement of those expected benefits (Amara et al. 2013; Wahba 2015).

The lack of studies that investigate the relationship between the degree of compliance with the disclosure requirements of financial instruments (IFRS 7) and board independence in GCC countries leads to the need to study this relationship. Additionally, the difference

in results from previous studies makes it difficult to predict a certain direction for this relationship. Accordingly, we hypothesize that:

Hypothesis 2 (H2). There is an association between board independence and the extent to which banks comply with IFRS 7 mandatory disclosure.

Board meeting frequency

Researchers consider that frequency of board meetings is a good indicator to increase the board's effectiveness. The more active a board is in conducting meetings, the more refined the supervisory functions of the board. This can also help to discuss the problems of the company, find appropriate solutions, and increase the efficiency of the company (Almaqtari 2019; Bagudo et al. 2018; Chen and Rezaee 2012; Ishak et al. 2017; Ojeka et al. 2019; Rouhou et al. 2015; Zango et al. 2015). These meetings reflect the role of the agent (management), and his keenness to discuss and monitor the issues of the company would improve performance-thus eliminating agency problems (agency theory) (Bae et al. 2018; Almaqtari 2019; Eluyela et al. 2018; Katmon and Farooque 2017). Several studies examine the impact of frequent board meetings on several aspects, such as earnings management (Katmon and Farooque 2017), accuracy of prediction (Karamanou and Vafeas 2005; Rouhou et al. 2015), audit report lag (Ahmed and Che-Ahmad 2016), the performance of banks and their progress (Eluyela et al. 2018; Khan et al. 2019), and quality of financial disclosure (Chen and Rezaee 2012; Ishak et al. 2017; Rouhou et al. 2015). With regard to compliance with IFRS, very few studies have addressed the effects of this factor. Abdullah (2011), Chen and Rezaee (2012), and Kent and Stewart (2008) find that board meetings contribute to increased compliance with regulations and IFRS disclosure requirements.

Despite the importance of holding regular meetings of the board of directors, there is some controversy about its shortcomings such as the effort and expense, plus it is time consuming (Almaqtari 2019). This, in turn, can affect the resources a company allocates toward increasing performance effectiveness, and may not rationalize the use of those resources properly (resource dependency theory). Additionally, it may not have a tangible effect, such as the study of Ojeka et al. (2019) which proves that there was no effect of the frequency of board meetings on compliance with the requirements of IFRS 7 in 14 listed banks in Nigeria. On the other hand, Bagudo et al. (2018) demonstrate that the frequency of board meetings has a negative impact on the degree of compliance with the disclosure requirements in IFRS. Based on the above, we hypothesize that:

Hypothesis 3 (H3). *There is an association between board meetings frequency and the extent to which banks comply with IFRS 7 mandatory disclosure.*

2.1. Internal Audit Committee (AC) Characteristics

2.1.1. Audit Committee Size

Audit committee is another important supplement to good governance. The efficiency of the audit committee may include the size of the committee, its independence, its existence, and its periodic meetings. In the literature, there is no agreement on the optimal number of audit committee members, but in general it is agreed that there should be at least three (Almaqtari 2019; Elzahar and Hussainey 2012; Juhmani 2017). From the standpoint of agency theory, it can be argued that the size of the audit committee may increase the level of internal control, integrity in work, the degree of compliance with laws, and solve various problems faced by the company (Bae et al. 2018). This, in turn, will reduce potential agency costs and problems. In addition, if the audit committee comprises a variety of experts who have high qualifications, this will increase the effectiveness of the committee and assist by increasing the monitoring system over financial reporting practices and transparency of disclosure (in the case of larger committee sizes). Another trend believes that with a large number of members who do not have the suitable requirements and great experience to deal with the work professionally will not benefit or support the committee's work, and

this may indirectly waste company resources (Almaqtari 2019; Bagudo et al. 2018; Juhmani 2017; Katmon and Farooque 2017; Krismiaji 2019).

Some studies tested the impact of audit committee size on different areas of a firm's work, such as earnings management (Katmon and Farooque 2017) and quality of risk disclosure (Elzahar and Hussainey 2012), and neither study found an impact. On the other hand, some researchers are interested in studying the impact of this factor on the extent of corporate disclosure, especially the level of compliance with IFRS. Abdullah (2011), Bagudo et al. (2018), Almaqtari (2019), and Krismiaji (2019) found a positive relationship between the size of the audit committee and the degree of compliance with IFRS. Conversely, Affes and Makni-Fourati (2019), Juhmani (2017), Agyei-Mensah (2019b), and Mnif and Znazen (2020) did not find any significant relationship between their variable and compliance with IFRS. Based on these discussions, we hypothesize that:

Hypothesis 4 (H4). *There is an association between audit committee size and the extent to which banks comply with IFRS 7 mandatory disclosure.*

2.1.2. Audit Committee Independence

While the audit committee is a committee of the board of directors, it contributes significantly to the efficiency of business supervision and risk management in an organization. In order for this committee to fulfill its responsibilities at the required level, it should be independent as far as possible from administrative pressures and be unbiased (Abdul Rahman and Hamdan 2017; Agyei-Mensah 2019a; Sellami and Fendri 2017). This independence ensures the members of the audit committee are unrelated to financial or personal relationships with the company or its executives, and therefore includes non-executive members (Ishak et al. 2017). In fact, the concept of independence and its application vary from one country to another, depending on each country's corporate governance code. One of the features of audit committee independence is to increase the level of control, which reduces the incidence of fraud and errors in work, raises the quality of financial reporting, and controls compliance with the required standards. All of this would be in line with the agency's concept, which reduces the gap between owners and management and increases transparency in disclosure (Almaqtari 2019; Juhmani 2017).

Some studies have shown that there is indeed a significant positive relationship between audit committee independence and the level of compliance with IFRS (Adznan and Nelson 2014; Agyei-Mensah 2019a; Juhmani 2017; Krismiaji 2019; Sellami and Fendri 2017; Mnif and Znazen 2020). However, there are also other studies that do not find any effect of this determinant over the degree of compliance, such as Abdul Abdul Rahman and Hamdan (2017), Agyei-Mensah (2017), Bagudo et al. (2018), and Almaqtari (2019). Given these conflicting results in the literature, we set the following hypothesis:

Hypothesis 5 (H5). *There is an association between audit committee independence and the extent to which banks comply with IFRS 7 mandatory disclosure.*

2.1.3. Audit Committee Meeting Frequency

According to the companies' law related to each country of the GCC, there is no specific number of such meetings that should be conducted by the audit committee, but the majority of countries suggest that meetings should take place at least four times a year (Shehata 2015). There is a clear link in the literature between the frequency of audit committee meetings and the effectiveness of its work. Committee meetings are considered one of the proxies of diligence that reflect the activity of the committee and are a good indicator of its keenness to provide the highest levels of supervision. Accordingly, the independent intelligent audit committee conducts regular meetings during the year and its members are keen to attend and discuss important and urgent problems. In addition, members meet in order to identify management risks and carry out the committee's responsibilities as required. Indeed, the topics discussed in the meetings include those

relating to financial reporting, preparation of financial statements, and quality of disclosure (Agyei-Mensah 2019a; Almaqtari 2019; Bagudo et al. 2018; Hoitash et al. 2009; Katmon and Farooque 2017; Sellami and Fendri 2017).

From the agency perspective, strengthening the level of the internal control system will support compliance with accounting standards (Bagudo et al. 2018). Although few studies have addressed this determinant, it has been shown to be a good determinant that helps to raise the level of disclosure and to increase compliance with IFRS (Ettredge et al. 2011; Alkurdi et al. 2019; Zango 2019). Conversely, several studies could not identify any significant association between the degree of compliance with IFRS and frequent audit committee meetings (Agyei-Mensah 2019a; Almaqtari 2019; Bagudo et al. 2018; Ernawati and Aryani 2019; Mnif and Znazen 2020; Sellami and Fendri 2017). Based on these discussions, we hypothesize that:

Hypothesis 6 (H6). *There is an association between audit committee meetings and the extent to which banks comply with IFRS 7 mandatory disclosure.*

2.2. Ownership Characteristics

2.2.1. Blockholder Ownership

Blockholder ownership refers to the concept of centralization of ownership (not dispersed), and they are usually the major shareholders (owning 5% or more). Through collecting the study data, it is clear that blockholder ownership is very common in the GCC countries, which is consistent with comments made by Grassa and Chakroun (2016). Blockholder ownership may contribute to reducing agency costs by supporting managers' control procedures and reducing differences in interests between owners and agents. They help to maintain their investments and improve the company's performance. However, they may cause the phenomenon of information asymmetry, because they will have priority in obtaining company information from internal sources. This could limit the disclosure of adequate and important information in the annual reports (Al-Hussain and Johnson 2009; Grassa and Chakroun 2016; Tessema 2019).

On the other hand, the proportion of major shareholders may be considered financially beneficial (finance theory)¹. They have a strong incentive to bear any financial burden for increasing the level of monitoring. They provide monitoring equivalent to owner control rather than regular managerial control. Thus, those shareholders provide a common public benefit: for the company, for the small shareholders, and for themselves as well. Accordingly, blockholder ownership would help to improve the quality of disclosure, increase transparency in annual reports, and raise the level of compliance with regulations and standards (Alfraih 2018).

The results of previous studies are varied in the issue of the relationship between blockholder ownership and the degree of compliance with standards and disclosure.

Grassa and Chakroun (2016) find that the proportion of major shareholders decreases the corporate governance disclosures in the GCC banks. Alfraih (2018) finds that there is a positive relationship between blockholder ownership and the disclosure of intellectual capital in Kuwait. Juhmani (2017), Abdul Abdul Rahman and Hamdan (2017), and Agyei-Mensah (2017); however, could not find any correlation between this factor and the level of compliance and disclosure in Bahrain, Malaysia and Ghana respectively. Consequently, we hypothesis that:

Hypothesis 7 (H7). There is an association between blockholder ownership and the extent to which banks comply with IFRS 7 mandatory disclosure.

2.2.2. Governmental Ownership

Governmental ownership is often expressed by the percentage of shares owned by the government in institutions. It might be one of the indicators of governance quality if it is joined by other factors related to the country's laws and business environment. It is also

considered a good governance mechanism if it meets the interests of the firm's investors and other stakeholders in satisfaction by complying with regulations and improving the quality of disclosure. This increases the confidence of stakeholders and maintains the 'legitimacy' of these organizations and their continuity. In addition, the companies that the government owns can have greater opportunities to obtain government financial funds, which helps to improve the companies' performance and compliance with accounting standards (Al-Akra et al. 2010; Alfraih and Almutawa 2017; Ebrahim and Fattah 2015). Alfraih and Almutawa (2017) found that governmental ownership increases the proportion of voluntary disclosure in Kuwaiti companies. Additionally, Elamer et al. (2019) found that government and family ownership improve the risk disclosure of banks in MENA countries.

On the other hand, there is another trend that believes that governmental ownership does not help in solving agency problems. While the government has the power to select the directors of the board, regardless of their qualifications and professional experience, the efficiency of the firm can be negatively affected. This creates the problem of monitoring those governmental directors and their compliance with the laws due to their lack of awareness of the importance of compliance in order to improve disclosure practices, in addition to individuals' confidence in the government-owned companies (of which they are implicit shareholders) which makes compliance costs higher than non-compliance. Moreover, the government can obtain the information it needs at any time directly from the company or indirectly from any other source, along with easy access to government financial funds, which weakens the corporate governance mechanism and the level of oversight. All of this leads to a weakening of pressure on companies owned by the government toward complying with mandatory disclosure and lowering the level of enforcement of laws and regulations toward their adherence to accounting standards (Alfraih 2018; Al-Janadi et al. 2013). Thus, Al-Janadi et al. (2013), Ebrahim and Abdel Ebrahim and Fattah (2015), and Alotaibi and Hussainey (2016) confirm this view by finding a negative relationship between the level of disclosure and governmental ownership in companies. In contrast, Al-Akra et al. (2010), Juhmani (2017), and Alfraih (2018) could not prove a relevant association between governmental ownership and the compliance level with IFRS 7.

Based on the above discussion, we hypothesize that:

Hypothesis 8 (H8). There is an association between governmental ownership and the extent to which banks comply with IFRS 7 mandatory disclosure.

2.2.3. Institutional Ownership

Institutional ownership is one part of the ownership structures that have been included in disclosure studies. This variable reflects the percentage of institutions owning shares in companies, which may affect their level of disclosure. From the point of view of agency theory, institutional shareholders assume companies comply with the rules and meet the required level of disclosure. Consequently, they are considered to be one of the corporate governance mechanisms that monitor and control the performance and discipline of managers and coordination between the interests of all stakeholders (Bae et al. 2018). This leads to a reduction in the agency's problems and an increase in the level of disclosure in the annual reports (Agyei-Mensah 2017; Alnabsha et al. 2018; Grassa and Chakroun 2016; Tessema 2019). A wide range of previous studies indicate that there is a positive relationship between both institutional ownership and financial disclosure. Khodadadi et al. (2010) find that institutional ownership increases the level of voluntary disclosure in Iran. Sarea and Al Sarea and Nesuf (2013) clarify that institutional ownership improves compliance with IAS 21 in Bahrain. Similarly, Elghaffar et al. (2019) found an increase in the level of risk disclosure in Egyptian banks, and Ebrahim and Abdel Ebrahim and Fattah (2015) an increase in the level of compliance with the requirements of IAS 12 in Egyptian companies. Moreover, Alnaas and Rashid (2019) demonstrate that compliance with IFRS in companies in North African countries (Morocco, Tunisia, Egypt) is positively affected by institutional ownership.

From an adverse viewpoint, institutional ownership may limit the degree of compliance with standards and the required level of disclosure as institutional investors may not care for the future vision and value of companies and, instead, are focusing on the present profits and positions (competing theory). They accordingly do not give much attention to monitoring managers and their behaviors or follow the work progress and systems in the company. Consequently, they have little inclination toward complying with the requirements of disclosure or even understanding the important future goals of the company (Tessema 2019). Collett and Dedman (2010) find a negative association between institutional ownership and the level of disclosure of share price movements by firms in the UK. Additionally, institutional ownership reduces the disclosure of internal control systems in EU firms (Michelon et al. 2009). Furthermore, some studies do not identify any important relationship between institutional ownership and level of disclosure, but the direction of the relationship suggests negativity between the two variables (Agyei-Mensah 2017; Al-Sartawi 2018; Yusuf et al. 2018). Therefore, we hypothesize that:

Hypothesis 9 (H9). *There is an association between institutional ownership and the extent to which banks comply with IFRS 7 mandatory disclosure.*

3. Research Design

3.1. Study Sample and Data Collection

The main focus of this study is to identify the effects of corporate governance (CG) determinants on the degree of compliance with IFRS 7. Thus, several criteria have been set in place to achieve the desired research objectives. First, the study sample includes the six GCC countries, which have a similar systemic and cultural base. The sample of the study includes 335 observations of 48 banks listed in the GCC markets from 2011 (the effective date of the last update of the standard by the IASB in 2010) until 2017 (the most recent data at the time of the study). Data collected from different sources including the official banks' websites, capital market websites from each country, Datastream; Bloomberg and banks' annual reports. Table 1 shows the number of observations in our study.

GCC	All Listed Banks	Not Meeting Criteria	Selected Banks	2011	2012	2013	2014	2015	2016	2017	Total Obs.
Saudi	12	0	12	12	12	12	12	12	12	12	84
Kuwait	12	2	10	9	10	10	10	10	10	10	69
Oman	8	2	6	6	6	6	6	6	6	6	42
Qatar	9	4	5	5	5	5	5	5	5	5	35
Bahrain	15	10	5	5	5	5	5	5	5	5	35
UAE	25	15	10	10	10	10	10	10	10	10	70
Total	81	33	48	47	48	48	48	48	48	48	335

Tal	ble	1.	Se	lected	d s	sam	ple.
-----	-----	----	----	--------	-----	-----	------

To investigate the different relationships between corporate governance determinants on one side and level of compliance with IFRS 7 disclosure requirements on the other, the following model has been employed:

$$\begin{split} D.Index_{it} &= \beta 0 + \beta_1 B.SIZE_{jt} + \beta_2 B.INDP_{jt} + \beta_3 B.MEET_{jt} + \beta_4 A.SIZE_{jt} + \beta_5 A.INDP_{jt} + \beta_6 A.MEET_{jt} + \beta_7 BK.OWN_{jt} + \beta_8 GV.OWN_{it} + \beta_9 IS.OWN_{it} + \sum Controls + \epsilon_{it} \end{split}$$

where:

- D.Index: the score of disclosure index
- B.SIZE: board size measured by the total number of directors on the board.
- B.INDP: board independence measured by the proportion of independent directors in the board.
- B.MEET: board meeting frequency measured by the number of board meetings per year.
- A.SIZE: audit committee size measured by the number of audit committee members.
- A.INDP: audit committee independence measured by the proportion of independent members in audit committee.
- A.MEET: audit committee meeting frequency measured by the number of meetings held by audit committee members per year.
- BK.OWN: blockholder ownership measured by the proportion of shares owned by substantial shareholders (5% or more).
- GV.OWN: governmental ownership measured by the proportion of shares owned by governmental institutions.
- IS.OWN: institutional ownership measured by the proportion of shares owned by financial institutions.
- Controls:
- firm size (F.SIZE): The natural logarithm of firms' total assets
- firm age (F.AGE): The number of years since foundation
- profitability (PROF): Return on equity: net income/shareholders' equity
- leverage (LEVR): Total liabilities/ shareholders' equity (debt to equity ratio)
- liquidity (LIQD): The total of loans to deposits
- complexity (CMPX): The number of subsidiaries of the firm in the year
- year dummies (YEARS)
- country dummies (COUNT).
- $\varepsilon = \text{Error term}$

3.1.2. The Dependent Variable (Compliance Index)

Literature on the compliance issue shows that the most common instrument employed in measuring the compliance level is by index (Al-Akra et al. 2010; Alsaeed 2006; Al-Shammari et al. 2008; Lopes and Rodrigues 2007). This index can be self-constructed or developed based on different sources, such as previous studies or trusted entities such as the indices used in Big Four auditing companies. Since this study measures the degree of compliance with the disclosure requirements of IFRS 7 for all listed GCC banks, we use the self-constructed index of Yamani and Hussainey (2021) which built based on the standard requirements, and it consists of 76 disclosure items. These items include the latest updates of the standard in 2010 and became effective mandatorily in 2011. The construction of this index is mainly based on IFRS 7 standard. The Cooke's method is used for scoring the index (dichotomous approach) due it being the most common method adopted in compliance research (Tsalavoutas et al. 2010) and its suitability for the purposes of the current study. In this method, we use unweighted dichotomous approach, where we give a disclosure item a score of 1 if this item is disclosed in the annual report and 0 otherwise. Table 2 shows our disclosure index (more details are given in Yamani and Hussainey 2021). Our compliance score is calculated as a ratio of the total of all applicable items disclosed by a company to the maximum possible number of disclosure items that should be disclosed (Equation (1)).

$$C_{j} = \frac{T = \sum_{i=1}^{n} d_{i}}{M = \sum_{i=1}^{m} d_{i}}$$
(1)

where (*Cj*) is the total compliance score for each bank (*j*), (*T*) is the total number of items disclosed (*di*) by bank (*j*), (*M*) is the maximum number of applicable disclosure items for bank (*j*) and (*di*) = 1 if item (*i*) is disclosed, 0 otherwise; m is the maximum number of items.

Table 2. The IFRS 7 disclosure index.

No.	Title
SIGNIFICANC	E OF FINANCIAL INSTRUMENTS FOR FINANCIAL POSITION AND PERFORMANCE
Statement of fir	nancial position
Categories of fi	nancial assets and financial liabilities
Carrying amoun	nts of each of the following categories shall be disclosed either in the statement of financial position or in the notes:
1.	Financial assets measured at fair value through profit or loss-designated
2.	Financial liabilities at fair value through profit or loss-designated
3.	Financial liabilities at fair value through profit or loss-held for trading
4.	Financial assets measured at amortized cost
5.	Financial liabilities measured at amortized cost
6.	Financial assets measured at fair value through other comprehensive income
7.	Investments in equity instruments at fair value through other comprehensive income-designated
	Reclassification
8.	Date and amount of reclassification
9.	Qualitative description of the change its effect on the entity's financial statements
	Offsetting financial assets and financial liabilities
10.	Offsetting disclosures information for all recognized financial instruments
	Collateral
11.	Financial assets pledged as collateral
12.	Terms and conditions relating to pledge
	Compound financial instruments with multiple embedded derivatives
13.	Instrument that contains both a liability and an equity component the instrument has multiple embedded derivatives
	Defaults and breaches
14.	Any defaults and breaches during the period of principal, interest, sinking fund, or redemption terms of those loans payable
	Statement of income: Items of income, expense, gains or losses—Other comprehensive income
15.	Net gains/losses on by classes of financial instruments at fair value (designated or held for trading)
16.	Net gains/losses on financial liabilities and financial assets measured at amortized cost
17.	Total interest revenue and total interest expense
18.	Fee income and expense
19.	Gain/loss arising from derecognition of financial assets measured at amortized cost

Table 2. Cont.

No.	Title
	Other disclosures
	Accounting policies
20.	Recognition and measurement for financial assets and financial liabilities designation
21.	The nature of financial assets measured at fair value through profit or loss-designated
22.	Terms and conditions for financial assets and financial liabilities designation
23.	Terms and conditions of impairment about financial instruments
	Hedge accounting
24.	An entity's risk management strategy and how it is applied to manage risk
25.	How the entity's hedging activities may affect the amount, timing and uncertainty of its future cash flows
26.	Description of the hedging instruments that are used to hedge risk exposures
27.	Gains/losses on hedge ineffectiveness associated with financial instrument
	Cash flow hedges (CFH)/hedge of a net investment in a foreign operation
28.	Forecast transaction for which hedge accounting had been used
29.	Carrying amount of the hedging instruments (financial assets separately from financial liabilities)
30.	The change in fair value of the hedging instrument used as the basis for recognizing hedge ineffectiveness for the period
31.	Gains/losses of CFH recognized in other comprehensive income
	Fair value hedges (FVH)
32.	Carrying amount of the hedging instruments (financial assets separately from financial liabilities)
33.	The change in fair value of the hedged item used as the basis for recognizing hedge ineffectiveness for the period
34.	Gains/losses of FVH
	Fair value
35.	Fair value for each class of financial assets and financial liabilities
36.	Comparable carrying amounts
37.	Measurement methods and assumptions
38.	Information if fair value cannot be recognized or measured
39.	Changes in fair value of financial instruments
	NATURE AND EXTENT OF RISKS ARISING FROM FINANCIAL INSTRUMENTS
	Credit risk
40.	Exposure to risk and how they arise—Qualitative information
41.	Objectives, policies and processes for managing the risk and the methods used to measure the risk
42.	Summary quantitative data: exposure to risk at the reporting date
43.	Concentrations of credit risk if not apparent from summary quantitative data and sensitivity analysis
44.	Amount of maximum exposure to credit risk (before deducting value collateral)
45.	A description of collateral held as security and other credit enhancements security and credit-impaired at the reporting date
46.	A summary of credit risk rating grades that shows credit quality of financial instruments by asset class
47.	Allowance account for credit losses—qualitative information
48.	Allowance account for credit losses—quantitative information (changes in the loss allowance during the period)

Table 2. Cont.

No.	Title
49.	Allowance account for credit losses—information about financial instruments for which credit-impaired/not credit-impaired
50.	Nature and carrying amount of assets obtained by taking possession of collateral it holds as security or calling on other credit enhancements
51.	Policies for disposing assets or use of it in its operations when the assets are not readily convertible into cash
	Liquidity risk
52.	Exposure to risk and how they arise—Qualitative information
53.	Objectives, policies and processes for managing the risk and the methods used to measure the risk
54.	Maturity analysis for financial liabilities that show the remaining contractual maturities
	Market risk—interest rate risk
55.	Exposure to risk and how they arise—Qualitative information
56.	Objectives, policies and processes for managing the risk and the methods used to measure the risk
57.	Summary quantitative data: exposure to risk at the reporting date
58.	Concentrations of interest rate risk if not apparent from summary quantitative data and sensitivity analysis
59.	Interest rate sensitivity analysis showing how profit or loss and equity would have been affected by changes in the relevant risk variable that were reasonably possible at that date
60.	Methods and assumptions used in preparing the sensitivity analysis
	Market risk—currency risk
61.	Exposure to risk and how they arise—Qualitative information
62.	Objectives, policies and processes for managing the risk and the methods used to measure the risk
63.	Summary quantitative data: exposure to risk at the reporting date
64.	Concentrations of currency risk if not apparent from summary quantitative data and sensitivity analysis
65.	Currency risk sensitivity analysis showing how profit or loss and equity would have been affected by changes in the relevant risk variable that were reasonably possible at that date
66.	Methods and assumptions used in preparing the sensitivity analysis
	Market risk—other price risk
67.	Exposure to risk and how they arise—Qualitative information
68.	Objectives, policies and processes for managing the risk and the methods used to measure the risk
69.	Summary quantitative data: exposure to risk at the reporting date
70.	Concentrations of other price risk if not apparent from summary quantitative data and sensitivity analysis
71.	Other price risk sensitivity analysis showing how profit or loss and equity would have been affected by changes in the relevant risk variable that were reasonably possible at that date
72.	Methods and assumptions used in preparing the sensitivity analysis
	TRANSFERS OF FINANCIAL ASSETS
73.	An entity shall provide the required disclosures for all transferred financial assets that are derecognition/not derecognized:
74.	The nature of the transferred financial assets
75.	The nature of the risks, rewards and liabilities associated with the transferred financial assets
76.	The carrying amounts of the transferred assets and the associated liabilities
77.	The gain or loss recognized at the date of transfer of the assets

Source: Yamani and Hussainey 2021.

3.1.3. The Independent Variables (Corporate Governance Determinants)

This study identifies the association between different corporate governance mechanisms and the compliance level with IFRS 7 in banks. Therefore, several corporate governance determinants were selected to achieve the desired results. These determinants are board size, board independence, board meeting frequency, audit committee size, audit committee independence, audit committee meeting frequency, blockholder ownership, governmental ownership, and institutional ownership.

3.1.4. The Control Variables

To examine the relationship between the dependent and independent variables, a range of control variables are included that would affect the dependent variable. This in order to address any potential endogeneity problems relating to omitted variables, we employ a set of firm-related factors to control for the studied associations (Wooldridge 2010). These control variables are selected based on the previous literature (Agyei-Mensah 2017; Gerged et al. 2021). These variables are bank size, profitability, leverage, liquidity, age, complexity, year dummies, and the country of domicile (country dummies).

4. Empirical Findings

Table 3 shows the descriptive analysis. It shows that none of the banks in the selected sample show full compliance with IFRS 7. The highest degree of compliance is 96%, and the lowest degree of compliance is 47%. The mean of the level of compliance for all banks is around 78.7%, with a standard deviation of 0.12 which shows that the data are symmetric around the mean of the variable.

Variable	Mean	Std. Dev.	Median	Min	Max
DIndex	0.787	0.12	0.80	0.47	0.96
BSIZE	9.2	1.375	9	5	12
BINDP	0.574	0.265	0.56	0	1
BMEET	7.036	2.501	6	2	12
ASIZE	3.651	0.789	3	2	5
AINDP	0.718	0.277	0.75	0	1
AMEET	5.645	1.842	5	2	11
BKOWN	0.531	0.207	0.54	0.05	0.903
GVOWN	0.259	0.197	0.22	0	0.793
ISOWN	0.376	0.223	0.36	0	0.891
FSize	32,480.243	36,637.583	20,554.98	794.1	245,639
FAGE	34.663	16.505	36	2	69
PROF	0.119	0.053	0.12	-0.156	0.242
LEVR	0.139	0.092	0.13	0	0.382
LIQD	0.956	0.181	0.94	0.471	1.449
CMPX	3.681	3.368	3	0	11

 Table 3. Descriptive Statistics.

The maximum board and audit committee sizes are 12 and 5, the minimum are 5 and 2, and the mean are 9 and 4, respectively. For the independence ratio, board and audit committee show similar results: for both the maximum is 100%, which is a good indicator of independence—however, both show non-independence (0) as a minimum ratio of the independence. In addition, both show means of over 50% of independence: board (57%) and audit committee (72%). For the frequency of board and audit committee meetings, they demonstrate similar results as well. The maximum number of meetings for the board is 12, the minimum is 2, and the mean is 7. Similarly, the largest number of meetings held by the audit committee during the year is 11, the lowest is 2, and the mean is 6 meetings. The mean blockholder ownership is 0.53, while the mean governmental ownership is 0.25. Institutional ownership has a percentage of 38% (mean). Both governmental and institutional ownership have zero as a minimum value for some banks.

The maximum value of assets (size) is 245,639 million USD, minimum asset value is 794.1, and the mean is 32,480.24. In addition, the age of the GCC banks included in the sample ranged between 2 and 69 years, with a mean of 35 years. Return on average equity, which reflects profitability, has a maximum value of 0.24.2, a minimum value of -0.156, a standard deviation value of 0.05. Compared with liquidity, leverage has the lowest average with 13.9%, and this means that leverage in the banks tends to be low. In contrast, liquidity reflects a good ratio with a mean of 0.956, a maximum of 1.440, a minimum of 0.471 and a standard deviation of 0.181. Finally, the maximum number of subsidiaries (complexity) of the banks is 11, the minimum is 0, the mean is 3.681, the standard deviation is 3.368.

Tables 4 and 5 show the correlation and the variance inflation factor (VIF) results for the dependent and independent variables. This will help checking whether there is any potential sign of Collinearity. The results suggest that multicollinearity does not appear to be a concern in explaining the regression results.

Table 6 shows the regression results of the full model that includes all explanatory variables. Based on the results of the running model 1, we find that all audit committee related variables and governmental ownership percentage are positively related with the compliance level with IFRS 7. We also find that board meeting is negatively associated with the compliance level with IFRS 7.

We use the Durbin–Wu–Hausman (DWH) test to detect the possible incidence of endogeneity issues of individual regressors. As our independent variables should not be correlated with the residuals. The findings of DWH tests suggest that the governance variables are exogenous rather than endogenous. Therefore, DWH tests suggest that endogeneity seemed not to be a major concern in the results of models and we can rely on the results of OLS.

Table 7 shows the results of running the analysis for the three categories of explanatory variables separately. The results are consistent with the results presented previously in the full model, except for the audit committee meeting, where we find a significant negative association with the compliance level with IFRS 7 when only including audit committee related variables.

Following this, we employ a two-step dynamic GMM regression model as a robustness check to address the potential endogeneity issue arising from reverse causality association between governance variables and compliance level with IFRS 7 and also to ensure that the findings of the full model and for each category models were not severely affected by the potential concerns of endogeneity problems (Blundell and Bond 1998). Table 8 shows the results of running GMM models for the three categories of explanatory variables separately.

lable 4. Pairwise correlations.																
Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)
(1) DIndex	1.000															
(2) BSIZE	0.176 *	1.000														
(3) BINDP	-0.157 *	-0.022	1.000													
(4) BMEET	-0.263 *	-0.031	0.245 *	1.000												
(5) ASIZE	0.245 *	0.338 *	-0.022	0.022	1.000											
(6) AINDP	0.219 *	0.222 *	0.121 *	-0.201 *	0.213 *	1.000										
(7) AMEET	-0.019	0.170 *	0.143 *	0.433 *	0.032	0.037	1.000									
(8) BKOWN	0.118 *	0.113 *	0.102	-0.052	0.238 *	0.203 *	-0.071	1.000								
(9) GVOWN	0.313 *	0.071	-0.292 *	-0.067	0.173 *	-0.116 *	-0.117 *	0.193 *	1.000							
(10) ISOWN	0.098	-0.028	-0.012	-0.162 *	-0.043	0.189 *	-0.049	0.710 *	0.018	1.000						
(11) FSize	0.031	0.043	-0.165 *	-0.159 *	0.084	0.021	0.069	0.062	0.214 *	0.176 *	1.000					
(12) FAGE	0.174 *	0.294 *	0.113 *	0.075	0.099	-0.030	0.148 *	0.030	-0.020	0.165 *	0.239 *	1.000				
(13) PROF	0.330 *	0.312 *	-0.080	-0.254 *	0.286 *	0.209 *	0.004	0.082	0.224 *	0.066	0.256 *	0.149 *	1.000			
(14) LEVR	-0.185 *	-0.261*	0.014	0.277 *	-0.209 *	-0.407 *	0.108 *	-0.034	-0.038	0.060	0.143 *	-0.021	-0.276*	1.000		
(15) LIQD	-0.244 *	-0.225 *	0.142 *	0.132 *	-0.175 *	-0.204 *	0.112 *	-0.015	-0.157 *	-0.040	0.080	-0.198 *	-0.254 *	0.589 *	1.000	
(16) CMPX	0.131 *	0.074	-0.256 *	-0.122 *	0.058	-0.104	0.059	-0.137 *	0.282 *	-0.047	0.567 *	0.073	0.154 *	0.109 *	0.013	1.000

Table 4. Pairwise correlations.

* p < 0.05.

15 of 23

	VIF	1/VIF
BKOWN	3.057	0.327
ISOWN	2.946	0.339
LEVR	2.068	0.483
FSize	1.843	0.543
LIQD	1.817	0.55
CMPX	1.715	0.583
BMEET	1.657	0.604
AINDP	1.473	0.679
BSIZE	1.457	0.686
GVOWN	1.443	0.693
FAGE	1.427	0.701
ASIZE	1.423	0.703
PROF	1.413	0.708
AMEET	1.403	0.713
BINDP	1.345	0.744
Mean VIF	1.766	

 Table 5. Variance inflation factor.

 Table 6. Regression results of full model.

	(1)	(2)	(3)
VARIABLES	OLS	2SLS	GMM
L. DIndex			0.546 ***
			(0.0903)
BSIZE	-0.00396	-0.00714	-0.00147
	(0.00497)	(0.00606)	(0.00246)
BINDP	-0.0251	-0.0253	0.00723
	(0.0246)	(0.0289)	(0.0139)
BMEET	-0.0119 ***	-0.0161 ***	0.000674
	(0.00297)	(0.00435)	(0.000972)
ASIZE	0.0198 **	0.0265 ***	-0.00311
	(0.00861)	(0.00915)	(0.00330)
AINDP	0.0792 ***	0.0685 **	5.37×10^{-5}
	(0.0248)	(0.0266)	(0.00869)
AMEET	0.00696 *	0.00936 **	0.00192 *
	(0.00367)	(0.00424)	(0.00112)
BKOWN	0.000762	-0.00297	0.000170
	(0.0476)	(0.0519)	(0.0311)
GVOWN	0.160 ***	0.160 ***	0.0378
	(0.0343)	(0.0366)	(0.0471)
ISOWN	-0.00125	0.00411	-0.0213
	(0.0436)	(0.0475)	(0.0289)
FSize	-0.0240	-0.0440*	-0.0382 **
	(0.0208)	(0.0229)	(0.0194)
FAGE	0.00134 ***	0.00155 ***	0.00295 ***
	(0.000429)	(0.000456)	(0.00110)
PROF	0.303 **	0.288 **	-0.0734 *
	(0.127)	(0.137)	(0.0390)
LEVR	0.104	0.121	0.0981 **
	(0.0886)	(0.0975)	(0.0440)
LIQD	-0.0701 *	-0.0649	-0.0437 **
	(0.0418)	(0.0469)	(0.0211)
CMPX	0.00144	0.00313	-0.00442 ***
	(0.00250)	(0.00263)	(0.00161)
Constant	0.779 ***	0.876 ***	0.478 ***
	(0.0960)	(0.114)	(0.0960)
Year dummies	Yes	Yes	Yes
Country dummies	Yes	Yes	Yes
Observations	335	287	287
R-squared	0.291	0.303	
Durbin (score) chi2(3) =		$1.20454 \ (p = 0.7519)$	
Wu-Hausman $F(3,268) =$		0.376514 (p = 0.7700)	

Standard errors in parentheses *** p < 0.01, ** p < 0.05, * p < 0.1.

	Board Characterstics (1)	Audit Committee Characterstics (2)	Ownership Structure (3)
VARIABLES			
BSIZE	0.00320		
	(0.00487)		
BINDP	-0.0368		
	(0.0246)		
BMEET	-0.00831 ***		
	(0.00271)		
FSIZELOG	0.00591	0.00505	0.0133
	(0.0207)	(0.0207)	(0.0206)
FAGE	0.000868 **	0.000746 *	0.000757 *
	(0.000433)	(0.000414)	(0.000405)
PROF	0.464 ***	0.458 ***	0.441 ***
	(0.131)	(0.129)	(0.127)
LEVR	0.00403	0.0431	-0.0952
	(0.0879)	(0.0890)	(0.0835)
LIQD	-0.0840 *	-0.0902 **	-0.0591
~	(0.0431)	(0.0425)	(0.0427)
CMPX	0.00115	0.00316	0.000221
	(0.00251)	(0.00250)	(0.00257)
ASIZE	, , , , , , , , , , , , , , , , , , ,	0.0183 **	× ,
		(0.00810)	
AINDP		0.0645 ***	
		(0.0247)	
AMEET		-0.0126 **	
		(0.00334)	
BKOWN		× /	0.00926
			(0.0439)
GVOWN			0.145 ***
			(0.0340)
ISOWN			0.0235
			(0.0405)
Constant	0.802 ***	0.655 ***	0.669 ***
	(0.0992)	(0.0845)	(0.0820)
Year dummies	Yes	Yes	Yes
Country dummies	Yes	Yes	Yes
R-squared	0.193	0.191	0.211

Table 7. Regression results of the three categories separately

The significance levels (two-tail test) are: * = 10%, ** = 5%, *** = 1%.

			_
Table 8. GMM models for the three cate	gories of expl	lanatory variables	s separately.

	Board Characteristics (1)	Audit Committee Characteristics (2)	Ownership Structure (3)
VARIABLES			
L.DIndex	0.572 ***	0.578 ***	0.554 ***
	(0.0919)	(0.0905)	(0.0889)
BSIZE	-0.00216		
	(0.00239)		
BINDP	0.00558		
	(0.0124)		
BMEET	-0.00117 **		
	(0.000947)		
FSIZELOG	-0.0350 *	-0.0391 **	-0.0413 **
	(0.0193)	(0.0187)	(0.0188)

	Board Characteristics (1)	Audit Committee Characteristics (2)	Ownership Structure (3)
FAGE	0.00276 **	0.00272 **	0.00293 ***
	(0.00110)	(0.00109)	(0.00109)
PROF	-0.0776 **	-0.0807 **	-0.101 ***
	(0.0383)	(0.0363)	(0.0360)
LEVR	0.118 ***	0.102 **	0.129 ***
	(0.0426)	(0.0433)	(0.0422)
LIQD	-0.0462 **	-0.0454 **	-0.0448 **
	(0.0210)	(0.0210)	(0.0209)
CMPX	-0.00451 ***	-0.00424 ***	-0.00452 ***
	(0.00159)	(0.00157)	(0.00159)
ASIZE		-0.00322	
		(0.00329)	
AINDP		-0.000790	
		(0.00775)	
AMEET		-0.00241 **	
		(0.00108)	
BKOWN			0.0231 **
			(0.0297)
GVOWN			0.0340
			(0.0461)
ISOWN			-0.0346
			(0.0279)
Constant	0.456 ***	0.462 ***	0.480 ***
	(0.0964)	(0.0946)	(0.0956)
Year dummies	Yes	Yes	Yes
Country dummies	Yes	Yes	Yes
Observations	287	287	287
Number of id	48	48	48

Table 8. Cont.

The significance levels (two-tail test) are: * = 10%, ** = 5%, *** = 1%.

5. Discussion

It seems that board meeting has a negative association with compliance level with IFRS 7. Even though most previous studies evidenced a positive impact of board size (Al-Akra et al. 2010; Alfraih 2016; Bagudo et al. 2018), Alfraih and Almutawa (2017) found a negative impact; that is, a small board size can enhance the commitment process with the standard regulation, and this also is consistent with agency theory (Bae et al. 2018). In addition, board independence confirms the perspective of stewardship theory, and shares the signs of this relationship with Adznan and Nelson (2014) and Elzahar et al. (2015) who find a negative sign of board independence, regardless of its significance. This indicates that companies in developing countries may tend more toward confidentiality and may conceal important information from independent directors, which hinders them from performing their responsibilities properly. Moreover, the independence of directors might be merely an external image (unreal) in order for companies to gain the trust and respect of shareholders. This, in turn, affects monitoring procedures and the efficiency of management to control compliance with accounting standards (Ebrahim and Fattah 2015). For board meeting frequency, Abdullah (2011) and Chen and Rezaee (2012) found a positive impact, unlike Bagudo et al. (2018) who found a negative effect of board meetings on compliance with IFRS in Nigeria. This result is in line with the perspective of the theory of resource dependence. In sum, the results only confirm the third hypothesis of the study (H3), and find a negative impact.

In addition, the results of the current study point to the important role that the audit committee plays in GCC banks, and its effective role in influencing the degree of compliance, especially with IFRS 7. This is because the size of the audit committee affects positively and significantly, in agreement with Abdullah (2011), Bagudo et al. (2018), Almaqtari (2019), and

Krismiaji (2019). Furthermore, the independence of the audit committee also increases the level of compliance with IFRS 7, consistent with Adznan and Nelson (2014), Juhmani (2017), Sellami and Fendri (2017), Agyei-Mensah (2019a), and Krismiaji (2019). Moreover, the frequency of the audit committee meetings also positively affects compliance, in agreement with studies conducted by Ettredge et al. (2011) and Alkurdi et al. (2019). These three variables (audit committee size, independence, meetings) are in line with agency theory in terms of improving the efficiency of the company's business and commitment to the standards, and the role they play in the Gulf banks. Hence, it can be said that the three study hypotheses (H4, H5, H6) are accepted.

For the ownership structure, only governmental ownership shows a significant positive impact on the level of compliance. Moreover, governmental ownership shows a significant positive association with IFRS 7 compliance. The concept of centralization of ownership and government participation in the GCC countries is widespread, which can make the government's role more effective in adhering to standards in these environments (agreeing with legitimacy theory). This result supports the findings of Elamer et al. (2019) who emphasize the effective role of governmental ownership to improve risk disclosure in MENA countries. For institutional ownership, due to the limited vision of the institutional investors for the near present in the GCC countries, they are not eager to improve compliance or follow specific procedures. Therefore, their competition to maximize profits in a short time can make institutional ownership one of the factors that impedes compliance with accounting standards. Therefore, it might be that institutional ownership does not improve compliance with IFRS 7 in the study sample—on the contrary, it limits this compliance. This negative direction agrees with several studies, such as Collett and Dedman (2010), Michelon et al. (2009), and Yusuf et al. (2018). Thus, it can be concluded that H8 is accepted. Lastly, some of the control variables show significant associations with the compliance level.

6. Conclusions

In this study, emphasis is placed on identifying the possible role of corporate governance (CG) determinants in the compliance level with IFRS 7. In total, 335 observations of the sample GCC banks were tested from 2011 to 2017. Three CG categories were tested: board characteristics, audit committee, and ownership, in addition to inserting a group of the banks' attributes (size, age, profitability, leverage, liquidity, complexity) as control variables. The findings show that only H3, H4, H5, H6, and H8 have been accepted. The board number of meetings has a negative impact on the compliance degree, consistent with resource dependence theory, respectively. In the same vein, governmental ownership shows a positive impact on the compliance degree. Furthermore, the other predictors (AC size, AC independence, number of AC meetings) have a positive impact on the compliance degree.

This study includes some practical implications on several parties: professional practitioners, auditors, and preparers of financial statements. They can improve compliance with IFRS in general and with IFRS 7 specifically by covering the required elements in the standards. Additionally, the results can help the regulatory and official authorities and policymakers in the country to develop regulations and intensify their efforts toward strengthening corporate governance systems, reviewing codes and following up their implementation. In addition, the findings provide more empirical evidence on the application of IFRS 7 from the reality of developing countries and from one of the most important sectors (the banking sector). This would support the efforts of the IASB toward accounting harmonization around the world.

We focus our analysis on IFRS 7 as a type of risk disclosure. Further research could consider testing the impact of governance on voluntary and mandatory risk disclosure. It would be interesting to examine whether certain governance mechanisms impact one more than the other. Although Elshandidy et al. (2013) examined the impact of board independence, ownership structure and audit environment on mandatory and voluntary

disclosure, researchers could contribute to disclosure literature by considering the impact of other governance mechanisms on voluntary and mandatory risk disclosure.

Author Contributions: Conceptualization, K.H. and A.Y.; methodology, K.H. and A.Y.; validation, K.H. and A.Y.; formal analysis, A.Y. and K.A.; investigation, A.Y.; resources, A.Y.; data curation, A.Y.; writing—original draft preparation, A.Y.; writing—review and editing, K.H. & K.A.; supervision, K.H. and K.A. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Informed Consent Statement: Not applicable.

Data Availability Statement: Data available on request due to restrictions.

Conflicts of Interest: The authors declare no conflict of interest.

Notes

¹ An aspect of finance theory discusses strategic planning associated with investors and owners' decisions. It trusts investor vision through the accounting bias of the real and expected value of the investment and the hedging strategies taken in the event of a risk. As accounting and finance are considered part of the economy, they address the aspect of understanding the operating and financing decisions of organisations. This theory provides a new lens for analysing corporate governance in institutions, especially those with a high concentration of ownership (Myers 1984; Negash 2013).

References

- Abdul Rahman, Azhar, and Mohd Diah Hamdan. 2017. The Extent of Compliance with FRS 101 Standard: Malaysian Evidence. *Journal of Applied Accounting Research* 18: 87–115. [CrossRef]
- Abdullah, Mazni. 2011. Compliance with International Financial Reporting Standards (IFRS) in a Developing Country: The Case of Malaysia. Stirling: University of Stirling.
- Adznan, Syaima, and Sherliza Nelson. 2014. Financial Instruments Disclosure Practices: Evidence from Malaysian Listed Firms. *Procedia-Social and Behavioral Sciences* 164: 62–67. [CrossRef]
- Affes, Houda, and Yosra Makni-Fourati. 2019. Compliance with IFRS 3, IAS 36 and IAS 38 by Canadian Companies: Economic Incentives or Opportunistic Choices? SSRN. [CrossRef]
- Agyei-Mensah, Ben Kwame. 2017. The Relationship between Corporate Governance Mechanisms and IFRS 7 Compliance: Evidence from an Emerging Market. *Corporate Governance (Bingley)* 17: 446–65. [CrossRef]
- Agyei-Mensah, Ben Kwame. 2019a. The Effect of Audit Committee Attributes on Compliance with IAS 24-Related Party Disclosure: An Empirical Study. *International Journal of Law and Management* 61: 266–85. [CrossRef]
- Agyei-Mensah, Ben Kwame. 2019b. IAS-38 Disclosure Compliance and Corporate Governance: Evidence from an Emerging Market. Corporate Governance: The International Journal of Business in Society 19: 419–37. [CrossRef]
- Ahmed, Mohammed Ishaq, and Ayoib Che-Ahmad. 2016. Effects of Corporate Governance Characteristics on Audit Report Lags. International Journal of Economics and Financial Issues 6: 159–64.
- Al Lawati, Hidaya Mustafa Ali, and Khaled Hussainey. 2021. Disclosure of forward-looking information: Does audit committee overlapping matter? *International Journal of Accounting and Performance Evaluation*. Unpulished Work.
- Al-Akra, Mahmoud, Ian A. Eddie, and Muhammad Jahangir Ali. 2010. The Influence of the Introduction of Accounting Disclosure Regulation on Mandatory Disclosure Compliance: Evidence from Jordan. *British Accounting Review* 42: 170–86. [CrossRef]
- Alanezi, Faisal S., and Saad S. Albuloushi. 2011. Does the Existence of Voluntary Audit Committees Really Affect IFRS-Required Disclosure the Kuwaiti Evidence. *International Journal of Disclosure and Governance* 8: 148–73. [CrossRef]
- Alanezi, Faisal S., Mishari M. Alfaraih, Eyad.A. Alrashaid, and Saad S. Albolushi. 2012. Dual/joint auditors and the level of compliance with international financial reporting standards (IFRS-required disclosure): The case of financial institutions in Kuwait'. *Journal of Economic and Administrative Sciences* 28: 109–29. [CrossRef]
- Alfaraih, Mishari. 2009. Compliance with International Financial Reporting Standards (IFRS) and the Value Relevance of Accounting Information in Emerging Stock Markets: Evidence from Kuwait. Brisbane: Queensland University of Technology, Available online: https://eprints.qut.edu.au/36377/ (accessed on 20 January 2020).
- Alfraih, Mishari M. 2016. The Effectiveness of Board of Directors' Characteristics in Mandatory Disclosure Compliance. *Journal of Financial Regulation and Compliance* 24: 154–76. [CrossRef]
- Alfraih, Mishari. 2018. The Role of Corporate Governance in Intellectual Capital Disclosure. *International Journal of Ethics and Systems* 34: 101–21. [CrossRef]
- Alfraih, Mishari M., and Abdullah M. Almutawa. 2017. Voluntary Disclosure and Corporate Governance: Empirical Evidence from Kuwait. *International Journal of Law and Management* 59: 217–36. [CrossRef]
- Al-Hussain, Adel, and Robert Johnson. 2009. Relationship between Corporate Governance Efficiency and Saudi Banks' Performance. *The Business Review* 14: 111–17.

- Al-Jabri, Hamdan, and Mujahid Hussain. 2012. Compliance with International Accounting Standards Requirements among Omani Listed Companies. Journal of Administrative and Economic Sciences, Qassim University 5: 75–105.
- Al-Janadi, Yaseen, Rashidah Abdul Rahman, and Normah Omar. 2013. Corporate Governance Mechanisms and Voluntary Disclosure in Saudi Arabia. *Research Journal of Finance and Accounting* 4: 25–35.
- Alkurdi, Amneh, Khaled Hussainey, Yasean Tahat, and Mohammad Aladwan. 2019. The impact of corporate governance on risk disclosure: Jordanian evidence. *Academy of Accounting and Financial Studies Journal* 23: 1–16.
- Allini, Alessandra, Luca Ferri, Marco Maffel, and Annamaria Zampella. 2019. 'Compliance to IFRS 7: Evidence from the Italian banking sector'. In Discourse, Communication and the Enterprise: Where Business Meets Discourse. Edited by Giuliana Elena Garzone and Giordano Walter. Newcastle upon Tyne: Cambridge Scholars Publishing, p. 298.
- Almaqtari, Abduljalil Gazem. 2019. Impact of Corporate Governance on IFRS Adoption: A Comparative Study of India and Some Selected Gulf Countries. Aligarh: Aligarh Muslim University.
- Al Mutawaa, Abudllah, and Aly Hewaidy. 2010. Disclosure Level and Compliance with IFRSs: An Empirical Investigation of Kuwaiti Companies. *The International Business & Economics Research Journal* 9: 33.
- Alnaas, Ali, and Afzalur Rashid. 2019. Firm Characteristics and Compliance with IAS/IFRS. *Journal of Financial Reporting and Accounting* 17: 383–410. [CrossRef]
- Alnabsha, Abdelrhman, Hussein Abdou, Collins Ntim, and Ahmed Elamer. 2018. Corporate Boards, Ownership Structures and Corporate Disclosures. *Journal of Applied Accounting Research* 19: 20–41. [CrossRef]
- Alotaibi, Khalid, and Khaled Hussainey. 2016. Determinants of CSR Disclosure Quantity and Quality: Evidence from Non-Financial Listed Firms in Saudi Arabia. *International Journal of Disclosure and Governance* 13: 364–93. [CrossRef]
- Alsaeed, Khalid. 2006. The Association between Firm-specific Characteristics and Disclosure. *Managerial Auditing Journal* 21: 476–96. [CrossRef]
- Al-Sartawi, Abdalmuttaleb Musleh. 2018. Does Institutional Ownership Affect the Level of Online Financial Disclosure? *Academy of Accounting and Financial Studies Journal* 22: 1–10.
- Al-Sartawi, Abdalmuttaleb M. A. Musleh, Fatema Alrawahi, and Zakeya Sanad. 2016. Corporate Governance and the Level of Compliance with International Accounting Standards (IAS-1): Evidence from Bahrain Bourse. *International Research Journal of Finance and Economics*, 110–22. Available online: http://www.internationalresearchjournaloffinanceandeconomics.com (accessed on 20 January 2020).
- Al-Shammari, Bader. 2011. Factors Influencing the Extent of Mandatory Compliance with International Financial Reporting Standards: The Case of Kuwaiti Listed Companies. *Journal of International Business and Economics* 11: 305–60.
- Al-Shammari, Bader, Philip Brown, and Ann Tarca. 2008. An Investigation of Compliance with International Accounting Standards by Listed Companies in the Gulf Co-Operation Council Member States. *International Journal of Accounting* 43: 425–47. [CrossRef]
- Amara, Ines, Anis Ben Amar, and Anis Jarboui. 2013. Detection of Fraud in Financial Statements: French Companies as a Case Study. International Journal of Academic Research in Accounting, Finance and Management Sciences 3: 40–51. [CrossRef]
- Bae, Seong Mi, Md Masud, Abdul Kaium, and Jong Dae Kim. 2018. Cross-Country Investigation of Corporate Governance and Corporate Sustainability Disclosure: A Signaling Theory Perspective. Sustainability 10: 2611. [CrossRef]
- Bagudo, Muhammad Mustapha, Rokiah Ishak, and Kamarul Bahrain Abdul Manaf. 2018. Corporate Governance Mechanisms and Compliance with International Financial Reporting Standards (IFRS) in Nigeria. National Academy of Managerial Staff of Culture and Arts Herald 1: 718–22.
- Blundell, Richard, and Stephen Bond. 1998. Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics* 87: 115–43. [CrossRef]
- Bova, Frencesco, and Raynolde Pereira. 2012. The Determinants and Consequences of Heterogeneous IFRS Compliance Levels Following Mandatory IFRS Adoption: Evidence from a Developing Country. *Journal of International Accounting Research* 11: 83–111. [CrossRef]
- Chen, Yu, and Zabihollah Rezaee. 2012. The Role of Corporate Governance in Convergence with IFRS: Evidence from China. International Journal of Accounting & Information Management 20: 171–88.
- Collett, Nick, and Elisabeth Dedman. 2010. Large Share Price Movements, the Disclosure of News and Corporate Governance Implications for Disclosure Rules. *Journal of Applied Accounting Research* 11: 109–32. [CrossRef]
- Ebrahim, Ahmed, and Tarek Abdel Fattah. 2015. Corporate Governance and Initial Compliance with IFRS in Emerging Markets: The Case of Income Tax Accounting in Egypt. *Journal of International Accounting, Auditing and Taxation* 24: 46–60. [CrossRef]
- Elamer, Ahmed A., Collins G. Ntim, Hussein A. Abdou, Alaa Mansour Zalata, and Mohamed Elmagrhi. 2019. The Impact of Multi-Layer Governance on Bank Risk Disclosure in Emerging Markets: The Case of Middle East and North Africa. Accounting Forum. [CrossRef]
- Elghaffar, Emad, Ahmed Abotalib, and Manal Khalil. 2019. Determining Factors That Affect Risk Disclosure Level in Egyptian Banks. Banks and Bank Systems 14: 15–171.
- Elshandidy, Tamer, Ian Fraser, and Khaled Hussainey. 2013. Aggregated, voluntary, and mandatory risk disclosures incentives: Evidence from UK FTSE All Share. *International Review of Financial Analysis* 30: 320–33. [CrossRef]
- Eluyela, Damilola, Olamide Akintimehin, Wisdom Okere, Emmanuel Ozordi, Godswill Osuma, Simon Ilogho, and Olufemi Oladipo. 2018. Datasets for Board Meeting Frequency and Financial Performance of Nigerian Deposit Money Banks. *Data in Brief* 19: 1852–55. [CrossRef] [PubMed]

- Elzahar, Hany, and Khaled Hussainey. 2012. Determinants of Narrative Risk Disclosures in UK Interim Reports. *The Journal of Risk Finance* 13: 133–47. [CrossRef]
- Elzahar, Hany, Khaled Hussainey, Francesco Mazzi, and Ioannis Tsalavoutas. 2015. Economic Consequences of Key Performance Indicators' Disclosure Quality. *International Review of Financial Analysis* 39: 96–112. [CrossRef]
- Ernawati, Dwi, and Anni Aryani. 2019. Controlling Shareholders, Audit Committee Characteristics, and Related Party Transaction Disclosure: Evidence from Indonesia. *Jurnal Keuangan Dan Perbankan* 23: 14–28. [CrossRef]
- Ettredge, Mike, Karla Johnstone, Mary Stone, and Qian Wang. 2011. The Effects of Firm Size, Corporate Governance Quality, and Bad News on Disclosure Compliance. *Review of Accounting Studies* 16: 866–89. [CrossRef]
- Gerged, Ali Meftah, Khaldoon Albitar, and Lara Al-Haddad. 2021. Corporate environmental disclosure and earnings management— The moderating role of corporate governance structures. *International Journal of Finance & Economics*. [CrossRef]
- Grassa, Rihab, and Raida Chakroun. 2016. Ownership Structure, Board's Characteristics and Corporate Governance Disclosure in GCC Banks: What about Islamic Banks? International Journal of Accounting, Auditing and Performance Evaluation 12: 360–95. [CrossRef]
- Hamdan, Allam Mohammed Mousa. 2020 'Audit committee characteristics and earnings conservatism in banking sector: Empirical study from GCC'. *Afro-Asian Journal of Finance and Accounting* 10: 1–23. [CrossRef]
- Hoitash, Udi, Rani Hoitash, and Jean Bedard. 2009. Corporate Governance and Internal Control over Financial Reporting: A Comparison of Regulatory Regimes. *The Accounting Review* 84: 839–68. [CrossRef]
- Huse, Morten. 2007. Boards, Governance and Value Creation: The Human Side of Corporate Governance. Cambridge: Cambridge University Press.
- Ishak, Rokiah, Noor Afza Amran, and Kamarul Bahrain Abdul Manaf. 2017. The Compliance Level amongst Malaysian Listed Companies Towards Corporate Governance Index. *International Journal of Economic Research* 14: 209–22.
- Istiningrum, Andian Ari. 2020. Corporate Governance, IFRS Disclosure, and Stock Liquidity in Indonesian Mining Companies. Advances in Economics, Business and Management Research 120: 271–78.
- Juhmani, Omar. 2017. Corporate Governance and the Level of Bahraini Corporate Compliance with IFRS Disclosure. *Journal of Applied* Accounting Research 18: 22–41. [CrossRef]
- Karamanou, Irene, and Nikos Vafeas. 2005. The Association between Corporate Boards, Audit Committees, and Management Earnings Forecasts: An Empirical Analysis. *Journal of Accounting Research* 43: 453–86. [CrossRef]
- Katmon, Nooraisah, and Omar Al Farooque. 2017. Exploring the Impact of Internal Corporate Governance on the Relation between Disclosure Quality and Earnings Management in the UK Listed Companies. *Journal of Business Ethics* 142: 345–67. [CrossRef]
- Kent, Pamela, and Jenny Stewart. 2008. Corporate Governance and Disclosures on the Transition to International Financial Reporting Standards. Accounting & Finance 48: 649–71.
- Khan, Faizan, Tarek Arman, and Bilal Eneizan. 2019. Impact of Corporate Governance Characteristics on Banks Performance. *International Journal of Academic Management Science Research* 3: 1–7.
- Khodadadi, Vali, Soheila Khazami, and Abbas Aflatooni. 2010. The Effect of Corporate Governance Structure on the Extent of Voluntary Disclosure in Iran. *Business Intelligence Journal* 3: 151–64.
- Krismiaji, Surifah. 2019. Corporate Governance and Firm's Compliance on Disclosure of International Financial Reporting Standards– Indonesian Evidence. International Journal of Accounting, Finance and Risk Management. 4: 24–33. [CrossRef]
- Lopes, Patrícia Teixeira, and Lúcia Lima Rodrigues. 2007. Accounting for Financial Instruments: An Analysis of the Determinants of Disclosure in the Portuguese Stock Exchange. *International Journal of Accounting* 42: 25–56. [CrossRef]
- Mbir, Deborah Esi Gyanba, Otuo Serebour Agyemang, George Tackie, and Mac Junior Abeka. 2020. IFRS Compliance, Corporate Governance and Financial Reporting Quality of GSE-Listed Non-Financial Firms. *Cogent Business & Management* 7: 1759856. [CrossRef]
- Melville, Alan. 2017. International Financial Reporting: A Practical Guide, 6th ed. London: Pearson Education Limited.
- Michelon, Giovanna, Sergio E. Beretta, and Saverio Bozzolan. 2009. Disclosure on Internal Control Systems as Substitute of Alternative Governance Mechanisms (February 2009). Available online: https://ssrn.com/abstract=1316323 (accessed on 20 January 2020).
- Mnif, Yousra, and Oumaima Znazen. 2020. Corporate Governance and Compliance with IFRS 7: The Case of Financial Institutions Listed in Canada. *Managerial Auditing Journal*. [CrossRef]
- Myers, Stewart C. 1984. Finance theory and financial strategy. Interfaces 14: 126–37. [CrossRef]
- Nahar, Shamsun, Mohammad Istiaq Azim, and Md Moazzem Hossain. 2020. Risk Disclosure and Risk Governance Characteristics: Evidence from a Developing Economy. *International Journal of Accounting & Information Management* 28: 577–605.
- Negash, Minga. 2013. Corporate governance and ownership structure: The case of Ethiopia. *Ethiopian E-Journal for Research and Innovation Foresight (Ee-JRIF)* 5: 33–50.
- Ojeka, Stephen, Alex Adegboye, Dorcas Adetula, Kofo Adegboye, and Inemesit Udoh. 2019. IFRS Adoption and CEO Compensation: Evidence from Listed Banks in Nigeria. *Banks and Bank Systems* 14: 1–8. [CrossRef]
- Rouhou, Nadia, Wyme Douagi, and Khaled Hussainey. 2015. The Effect of IFRS Enforcement Factors on Analysts' Earnings Forecasts Accuracy. Corporate Ownership and Control 13: 256–81. [CrossRef]
- Sanni, Mubaraq, Ismaila Daudu Aliu, and Yemisi Esther Olanrewaju. 2020. Board Characteristics and International Financial Reporting Standards (IFRS) Compliance among Nigerian Listed Companies: Mixed Method Approach. Global Journal of Accounting 6: 24–40.

- Santos, Edilene Santana, Vera Maria Rodrigues Ponte, and Patricia Vasconcelos Rocha Mapurunga. 2013. Mandatory IFRS Adoption in Brazil (2010): Index of Compliance with Disclosure Requirements and Explanatory Factors of Firms Reporting. VII Congresso Da Associação Nacional Dos Cursos de Pós-Graduação Em Ciências Contábeis—ANPCONT. 1–37. [CrossRef]
- Sarea, Adel Mohammed, and Hesa Jasim Al Nesuf. 2013. The Impact of Management Structure and Bank Characteristics on the Level of Compliance with IAS 21: Evidence from Bahrain. *International Journal of Accounting and Taxation* 1: 58–68.
- Sellami, Yosra, and Hela Fendri. 2017. The Effect of Audit Committee Characteristics on Compliance with IFRS for Related Party Disclosures: Evidence from South Africa. *Managerial Auditing Journal* 32: 603–26. [CrossRef]
- Shehata, Nermeen. 2015. Development of Corporate Governance Codes in the GCC: An Overview. *Corporate Governance* 15: 315–38. [CrossRef]
- Swedan, Ziyad, and Sohail Ahmed. 2019. The Impact of Corporate Governance Implementation on Firm's Performance in the Gulf Countries (GCC) Based on Stewardship Assumptions. *International Journal of Business Society* 2: 42–47.
- Tahat, Yaseen, Ghassan Mardini, and David Power. 2017. Factors Affecting Financial Instruments Disclosure in Emerging Economies: The Case of Jordan. *Afro-Asian Journal of Finance and Accounting* 7. [CrossRef]
- Tauringana, Venancio, and Lyton Chithambo. 2016. Determinants of Risk Disclosure Compliance in Malawi: A Mixed-Method Approach. *Journal of Accounting in Emerging Economies* 6: 111–37. [CrossRef]
- Tessema, Abiot. 2019. The Impact of Corporate Governance and Political Connections on Information Asymmetry: International Evidence from Banks in the Gulf Cooperation Council Member Countries. *Journal of International Accounting, Auditing and Taxation* 35: 1–17. [CrossRef]
- Tsalavoutas, Ioannis, Lisa Evans, and Mike Smith. 2010. Comparison of Two Methods for Measuring Compliance with IFRS Mandatory Disclosure Requirements. *Journal of Applied Accounting Research* 11: 213–28. [CrossRef]
- Tsalavoutas, Ioannis. 2011. Transition to IFRS and Compliance with Mandatory Disclosure Requirements: What Is the Signal? Advances in Accounting 27: 390–405. [CrossRef]
- Uyar, Ali, Merve Kılıç, and Başak Ataman Gökçen. 2016. Compliance with IAS/IFRS and Firm Characteristics: Evidence from the Emerging Capital Market of Turkey. *Economic Research-Ekonomska Istrazivanja* 29: 148–61. [CrossRef]
- Wahba, Hayam. 2015. The Joint Effect of Board Characteristics on Financial Performance. *Review of Accounting and Finance* 14: 20–40. [CrossRef]
- Wooldridge, Jeffrey. 2010. Econometric Analysis of Cross Section and Panel Data. New York: MIT Press.
- Yamani, Amal, and Khaled Hussainey. 2021. Compliance with IFRS 7 by financial Institutions: Evidence from GCC. International Journal of Disclosure and Governance 18: 42–57. [CrossRef]
- Yusuf, Mohammed, Musa Fodio, and Murine Nwala. 2018. Effect of Ownership Structure on Voluntary Disclosure of Listed Financial Firms in Nigeria. *International Journal of Economics, Commerce And Management* 6: 493–516.
- Zango, Adamu. 2019. Monitoring Mechanisms and Risk Disclosure of Banks in Nigeria a Contextual Synopsis. Uluslararası Sosyal Bilimler Dergisi 3: 144–63.
- Zango, Adamu, Hasnah Kamardin, and Rokiah Ishak. 2015. Corporate Governance and IFRS 7 Disclosure in Nigerian Banks: A Triple Model Assessment. Paper present at the International Conference on Accounting Studies (ICAS) 2015, Johor Bahru, Malaysia, August 17–20.